Author’s response to reviews

Title: An evaluative study of the benefits of participating in intergenerational playgroups in aged care for older people

Authors:

Margaret C Skropeta Dr (m.skropeta@uws.edu.au)
Alf C Colvin Dr (a.colvin@uws.edu.au)
Shannon Sladen Ms (ssladen@unitingcareswact.org.au)

Version: 5 Date: 17 September 2014

Author’s response to reviews: see over
Dear Editor,

Thankyou for your comments, we have a statement within the manuscript regarding how this manuscript adheres to the RATS guidelines to comply with BMC guidelines. It is in the Methodological considerations section and is on page 24 as it addresses the important qualitative methodological issues in relation to this paper. It is highlighted in yellow.

Kind regards

Dr C Margaret Skropeta
PhD, MA (Hons), BA
Senior Lecturer (Therapeutic Recreation)
School of Science and Health
Bachelor of Health Science
Campbelltown Campus, 24.4.25

Ph: 46203232
Email: m.skropeta@uws.edu.au